Author’s response to reviews

Title: Multiple Intraneural Glomus Tumors In Different Digital Nerve Fascicles

Authors:
Ying Wang (1875732913@qq.com)
Hui Lu (huilu@zju.edu.cn; hitman1982@hotmail.com)

Version: 3 Date: 24 Aug 2019

Author’s response to reviews:

Dear Prof. Catherine Rice:

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. The response to your comments are listed below this letter.

We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Yours sincerely, Hui Lu M.D. Department of orthopedics

, The First Affiliated Hospital, College of Medicine, ZheJiang University. Huilu@zju.edu.cn
Replies to the Editor Comments:

Thank you for editing your manuscript according to our requests. However, there still remain some minor concerns which need to be addressed before we can provisionally accept the manuscript for publication

1- Please label all figure panels clearly and according to figure legend; in figure 4 it is not clear which of the images is which. Please label them A to I to correspond to the figure legend.

Please do the same for the supplementary file figure where A- E correspond to legend details: (A)CK(-), (B)C31(-), (C)Desmin(-), (D)EMA(-) and (E)CgA(-).

Answer: we relabel the figure 4 and figure S.

2- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Answer: yes, we confirm it.